Author's response to reviews

Title: A markov model to evaluate hospital readmission

Authors:

Nicola Bartolomeo (nicolabartolomeo@virgilio.it)
Paolo Trerotoli (trerotoli@igiene.uniba.it)
Annamaria Moretti (moretti@qubisoft.it)
Gabriella Serio (seriogab@igiene.uniba.it)

Version: 4 Date: 21 April 2008

Author’s response to reviews:

Dear editor,
we have revised the manuscript and we have made other correction. Specifically relating to your request:
author list: we have modified the list and now name in the submission sistem and on the paper are identical
astesrixes have been removed, because they indicate participation to the paper and it is specified in the appropriate section
country have been specified
keywords have been removed
competing interest: the word have been corrected
Figure legends: the word "legends" have been added
Figure cropping: figure have been cropped more than the first revision.
Best wishes
Paolo Trerotoli